Author's response to reviews

Title: Selective embolization for hypervascular metastasis from differentiated thyroid cancer: report of two cases

Authors:

Hee Young Son (hyson79@gmail.com)
Soo-Youn An (ent@dirams.re.kr)
Eui Young Kim (euiykim@dirams.re.kr)
Sang Bu Ahn (sangbu@dirams.re.kr)
Byung Chul Lee (bleemdir@gmail.com)

Version:3 Date:14 August 2014

Author's response to reviews: see over
August 14, 2014

Journal of Medical Case Reports
c/o BioMed Central
236 Gray’s Inn Road
London WC1X 8HB
United Kingdom
info@biomedcentral.com / tel: +44 (0) 20 3192 2009 / fax: +44 (0)20 3192 2010

Dear Professor. Michael Kidd AM
Enclosed please find a case report entitled, “Selective embolization for hypervascular metastasis from differentiated thyroid cancer: report of two cases” by Son et al., which we wish to submit for publication in the “Journal of Medical Case Reports”.

This is a epoch-making report to document the effectiveness of embolization for an immaculate removal of skeletal metastases from thyroid malignancy. This study is cause for changes in the existing therapies of thyroid malignancy.

Your kind consideration of this paper would be greatly appreciated. These data have not reported elsewhere. This manuscript has not been submitted elsewhere. All authors took part in the work and agree with the contents of the manuscripts.

Sincerely yours,

Hee Young Son, M.D.

Hee Young Son, M.D.